Author's response to reviews

**Title:** Effect of questionnaire length, personalisation and reminder type on response rate to a complex postal survey: a randomised controlled trial

**Authors:**

- Shannon Sahlqvist (slf45@medschl.cam.ac.uk)
- Yena Song (Y.Song@soton.ac.uk)
- Fiona Bull (F.C.L.Bull@lboro.ac.uk)
- Emma Adams (E.J.Adams@lboro.ac.uk)
- John Preston (jpreston@soton.ac.uk)
- David Ogilvie (david.ogilvie@mrc-epid.cam.ac.uk)

**Version:** 3  **Date:** 21 April 2011

**Author's response to reviews:** see over
Dear Editors,

Re: Effect of questionnaire length, personalisation and reminder type on response rate to a complex postal survey: a randomised controlled trial (MS: 1062463551506286)

Thank you for agreeing to publish this manuscript. I have carefully proof-read the manuscript and addressed the comments made. Several minor textual changes were made to the manuscript and they are as follows:

- Line numbers have been removed
- Text that had been revised was underlined and the underlining has been removed
- A mistake was identified in Figure 1 (a cell read ‘1’ instead of ‘15’) and this has been amended.

Thank you once again for publishing this manuscript.

Yours sincerely,

Shannon Sahlqvist